Author's response to reviews

Title: How are health professionals earning their living in Malawi?

Authors:

Adamson S Muula (muula@email.unc.edu)
Fresier C Maseko (fcmaseko@yahoo.com)

Version: 4 Date: 9 August 2006

Author's response to reviews: see over
9th August 2006

Dear Editor,

The following changes were suggested and we have reformatted the manuscript.

Affiliations - Each affiliation to be written in full, in the following format: department, institute, city, and country. **We have been able to do this.**

Abstract - Set your conclusions section heading in bold. We have done this.

We have renamed Materials & Methods – to 'Methods'.

We have inserted Conclusions section after Discussions

We have spaced the Acknowledgements section heading in the same style as all other section headings.

We have placed the appendices before the Acknowledgements section.

Thank you.

Adamson S. Muula
Corresponding author